Dear editors,

We sincerely thank the editor and all reviewers for their valuable feedback that we have used to improve the quality of our manuscript. We revised the manuscript according to the comments point-by-point. All the revises are listed as below.

1. Modified conclusion part as the reviewer’s suggestion.
2. As all the two reviewers suggested to adding the information prior to the patient were received, we collect information of the patient in another hospital before she was transferred to our department. All information we got has been added in the "History of present illness" part.
3. We significantly enrich the part of discussion according to the reviewers and editor’s suggestion. We refer the clinical and scientific significance of our case report and point out how it can help in clinical practice.
4. We expand reference from 13 pieces to 20 pieces, which are mostly published in recent years. These references are close to our finding in case.
5. We modify the references section to meet the format for references guidelines of WJCC.
6. We provide an English level certificate from American Journal Experts, as the recommendation of WJCC website.

Once again, thank you very much for your constructive comments and suggestions, which would help us to improve the quality of the manuscript.